Author's response to reviews

Title: Acute Fatal Posthypoxic Leukoencephalopathy Following Benzodiazepine Overdose - A Case Report and Review of the Literature

Authors:

Fawaz Al-hussain (faalhussain@ksu.edu.sa)
Salman Aljarallah (saljarallah@ksu.edu.sa)

Version: 2 Date: 18 August 2014

Author's response to reviews:

We made the following changes to the manuscript:

1) Regarding the consent, it was actually obtained from the father as you might have guessed. The “son” was written by mistake.

2) It was unclear to me what exactly you meant by “adhering to the formatting policy” as I followed all the instructions in the link you provided.

Nevertheless, I made the following modifications:

a) The title format was changed from capitalizing the first letter in each word to small letter except the first letter of the first word.

b) We moved the subheadings “Discussion, Post-hypoxic Leukoencephalopathy, Introduction, Clinical presentation of DPHL, Laboratory findings in DPHL, Imaging findings in DPHL, Pathologic findings in DPHL, Treatment of DPHL, Prognosis” to be under Conclusions.

I hope this answers your questions and helps in publishing and we are ready to change anything as you suggest.

Salman Aljarallah